Author's response to reviews

Title: Actigraphy as a diagnostic aid for REM sleep behavior disorder in Parkinson's disease

Authors:

Maartje Louter (maartje.louter@gmail.com)
Johan B.A.M. Arends (arendsj@kempenhaeghe.nl)
Bastiaan R. Bloem (b.bloem@neuro.umcn.nl)
Sebastiaan Overeem (s.overeem@neuro.umcn.nl)

Version: 4 Date: 24 March 2014

Author's response to reviews: see over
Dear Dr. Bhidayasiri,

Thank you for accepting our manuscript “Actigraphy as a diagnostic aid for REM sleep behavior disorder in Parkinson’s disease”.

We have now changed the minor revision of reviewer 3.

With kind regards, also on behalf of the other authors,

Sebastiaan Overeem

Sebastiaan Overeem, MD, PhD
Department of Neurology, Donders Institute for Brain, Cognition and Behavior
Radboud University Nijmegen Medical Centre
PO Box 9101, 6500 HB Nijmegen, The Netherlands
Tel: +31 24 361 52 02, Email: S.Overeem@neuro.umcn.nl
Comments of reviewer 3:
Page 7, paragraph 2: "As several clinical characteristics differed between patients with and without PD, comparisons were repeated using multiple regression analyses corrected for these variables, however this did not change the results.

This has now been changed in the manuscript."